



Article Exploring Safety Culture in Urban Air Mobility: System of Systems Perspectives Using Enterprise Architecture

Raquel Hoffmann^{1,*}, Hidekazu Nishimura¹, and Paulo Gomes²

- ¹ System Design and Management Graduate School, Keio University, Kanagawa 223-8526, Japan; h.nishimura@sdm.keio.ac.jp
- ² School of Aviation, Faculty of Engineering, Lund University, 246 51 Lund, Sweden; paulocgo@hotmail.com
- Correspondence: raquel.hoffmann@keio.jp

Abstract: Urban Air Mobility (UAM) is an emerging industry marked by technological advancements, new operational contexts, and regulatory frameworks. This article examines how to improve safety management in UAM operations by adopting a just culture approach from a system of systems perspective. Acknowledging the critical role of front-line workers, especially in the early stage with piloted vehicles, the ecosystem-level approach comprehends multiple providers, operators, and services. Employing an enterprise architecture methodology, we address the challenge of fostering a learning-oriented environment amidst diverse organizational perspectives and stakeholders' interests. This study identifies key capabilities, functions, and resource exchanges within and across organizations by strategically leveraging architectural views and systemic visualizations. A unified safety committee is discussed and recommended to facilitate consensus among stakeholders, including regulatory bodies, thus paving the way for industry-wide improvements. Findings contribute to evolving safety protocols in UAM operations and serve as a blueprint for integrating cutting-edge methodologies to drive systemic enhancements.

Keywords: safety culture; just culture; enterprise architecture; unified architecture framework (UAF); urban air mobility (UAM)



systems12050178

Received: 29 April 2024

Revised: 13 May 2024

Accepted: 15 May 2024

Published: 17 May 2024

(†)

(cc

Citation: Hoffmann, R.; Nishimura, H.; Gomes, P. Exploring Safety

System of Systems Perspectives Using

Enterprise Architecture. Systems 2024,

12, 178. https://doi.org/10.3390/

Academic Editor: Mahyar Amirgholy

Copyright: © 2024 by the authors.

Licensee MDPI, Basel, Switzerland.

This article is an open access article

distributed under the terms and

conditions of the Creative Commons

Attribution (CC BY) license (https://

creativecommons.org/licenses/by/

Culture in Urban Air Mobility:

1. Introduction

The International Civil Aviation Organisation (ICAO) defines safety in aviation as the "state in which risks associated with aviation activities, related to, or in direct support of the operation of aircraft, are reduced and controlled to an acceptable level". Managing risks and maintaining safety at an acceptable level in a complex and dynamic context requires a proactive and ongoing process. A safety management system (SMS) is a systematic approach to managing safety within aviation organizations. It encompasses the organizational structure, policies, procedures, and practices that enable an organization to effectively identify, assess, and mitigate risks to ensure the highest level of safety [1].

Safety culture refers to the shared values, attitudes, beliefs, and behaviors within an organization or industry that influence safety-related decisions and actions. It is the product of individual and collective attitudes toward safety and the organization's commitment to fostering a safe environment [2–4]. Safety management and safety culture are closely intertwined and complementary. An effective safety management system provides the framework and tools for identifying, assessing, and mitigating risks. Its success depends on the flow of information between different levels and how the organization responds to accidents and incidents [5].

Urban Air Mobility (UAM) envisions a flight operation in and around urban areas motivated by the challenges of congested urban scenarios. By utilizing vertical takeoff and landing (VTOL) aircraft, UAM expects to provide efficient and time-saving transportation options for commuters. Some UAM concepts of operations have been defined as reuniting stakeholders' needs and evolving the discussion around implementing UAM operations [6–8]. The challenges of implementing a complex and safety-critical operation like UAM transportation are significant. Effective solutions for safe transportation will affect the public acceptance and the sustainability of UAM operations [9–11].

Given the complexity and potential risks associated with operating aerial vehicles in densely populated urban environments, cultivating a just culture is crucial for encouraging open communication and reporting safety concerns [5]. It allows stakeholders, including pilots, operators, regulators, and the public, to collaborate effectively in identifying hazards, analyzing incidents, and implementing corrective actions to enhance safety standards [12]. Additionally, UAM operations are conceptual and planned to occur in the near future [6]. Unlike traditional aviation systems that have matured over decades, UAM represents a paradigm shift with no prior operational experience to draw upon. As a result, the emerging industry needs to be prepared for real-world operations, learning from mistakes to refine operational procedures, establish best practices, and ensure the highest safety standards [9,11,13,14].

A strong safety culture, with appropriate safety regulations for the UAM context, can integrate safety considerations into all aspects of UAM operations, from design and development to implementation and ongoing management. It instills confidence among passengers and stakeholders, promotes regulatory compliance, and ultimately contributes to the long-term sustainability and success of UAM as a safe and reliable mode of transportation in urban settings [14]. Consequently, cultivating a robust safety culture becomes paramount for successfully implementing UAM [15,16]. Establishing a just culture within the UAM ecosystem can promote collective responsibility for safety, fostering a mindset that continually prioritizes learning from incidents and near-misses to improve operational practices.

This study aims to understand the structure and perspectives of the safety management that would enable a just culture in the UAM operations. To this end, this study proposes organizational views for enabling a just culture in the UAM ecosystem using enterprise modeling. First, the safety management problem space was analyzed, and boundaries and drivers were identified according to the stakeholders' interests. Then, the envisioned solutions were represented in organizational views using enterprise modeling. The research questions intended to be answered in this article are the following:

RQ1: How does the organization balance stakeholders with different interests to achieve trust and share risk information? Refer to the results in Section 4.2.2.

RQ2: What processes and relationships support values like trust, ownership, and accountability to the front-line worker as a potential reporter? Refer to the results in Section 4.2.3. RQ3: How can we achieve a consensus on acceptable and unacceptable behavior among organizations of the UAM ecosystem? Refer to Section 5.

This research contributes to the UAM field by offering enterprise architecture views and insights into fostering a just culture, ultimately enhancing operational learning and overall safety performance. This study was conducted with a multidisciplinary team comprising experienced systems engineers with a background in aviation and safety, a senior systems engineering researcher, and a safety researcher and practitioner. Safety researchers provided theoretical foundations, while experts in modeling techniques discussed and refined the architectural views presented in this article. This collaborative effort involved a series of iterations and discussions to enhance the clarity and effectiveness of the proposed solutions. Subsequently, the work was presented to systems engineers at different hierarchical levels to gather valuable feedback and insights. Additionally, validation exercises were carried out with commercial pilots and safety practitioners to ensure the practical applicability and relevance of the proposed solutions.

The process of understanding the problem and proposing solutions was recursive and iterative in nature. Modeling, particularly in complex domains like aviation safety and system engineering, often involves implicit knowledge that may not be readily apparent. Therefore, verifying the clarity and usefulness of ideas through peer review and feedback from industry experts is critical to ensure that the proposed solutions meet the expected standards and effectively address real-world challenges. This iterative approach not only refines the conceptual models but also enhances the overall quality and applicability of the solutions proposed in this study.

The structure of this article is outlined in Figure 1. The rest of this article is organized as follows. Section 2 describes the theoretical foundations of this study for safety culture, just culture, and modeling. Then, contextual information about safety management in aviation and UAM operations is presented. Section 3 introduces the study method by explaining the need for a systemic approach, describing the enterprise architecture (EA) methodology and the modeling framework used to elaborate the architecture views. Section 4 presents the results of the study. First is the conceptualization process, which includes problem space definition and solution envisioning. Next, architectural views with safety management perspectives and definitions are presented. Section 5 debates the results, reflecting on the literature and proposing a cooperation framework for just culture at the ecosystem level. Limitations and research guidance for future contributions complete the discussion section. Finally, Section 6 contains conclusions.



Figure 1. Structure of this article.

2. Background

The systemic approach to address safety culture in the upcoming UAM operation requires an interdisciplinary study. The background section is composed of theoretical foundations and contextual operational data. First, we present the rationale for the systemic approach to address the problem. Then, theoretical foundations of safety (resilience, culture, and organizational aspects) and modeling (for UAM ecosystem representation) are introduced. Next, data about the aviation domain are presented in two topics: safety management regarding industry standards and regulation and the UAM Concept of Operations. The contextual operational data are built from existing practices of the commercial aviation field and information published by authorities, the eVTOL industry, and academia. These background topics are essential for defining the problem space and proposing the solution presented in the results (Section 4).

2.1. Need for Systemic Approach

Managing risks in critical safety operations, like air passenger transportation, combines different perspectives, roles, behaviors, and processes. Similarly, SMS should encompass the entire organization comprehensively, considering elements from top to bottom, including communication channels, personnel, and more. Additionally, it should account for the 'environment', encompassing all external circumstances that impact the system and necessitate a response, such as political and economic influences [17]. A systemic approach is essential, defined as the effort to perceive things holistically, viewing events, even failures, as outcomes of a system's operation. ICAO well defines the need to apply a systemic safety management approach with the aviation industry's SMS framework [1].

A just culture depends on a healthy safety culture and a supportive safety management system. Both include human operators (with their behaviors, beliefs, subjectivity, and personal aspects), organizational structures, management styles and rules, policies, technical processes, and human-made systems. Systematic evaluations demonstrated value in understanding the relationship between employee learning and organizational supportive conditions [18,19]. Moreover, a systemic approach is suggested for managing the paradoxical states and balancing organizational dynamics [20].

2.2. Safety Culture and Just Culture: Theoretical Background

ICAO defines Safety culture as "the enduring value, priority, and commitment placed on safety by every individual and every group at every level of the organization. Safety culture reflects the individual, group, and organizational attitudes, norms, and behaviors related to the safe provision of air navigation services." [21]. According to Reason [2], safety culture comprises five elements: informed culture, flexible culture, reporting culture, just culture, and learning culture. These elements are interconnected and interdependent. For instance, the presence of an informed culture relies on a robust reporting culture, which, in turn, depends on implementing a just culture.

In the aviation industry, there has traditionally been a prevailing belief that a greater likelihood of assigning blame accompanies higher levels of professional responsibility. In many countries, pilots receive significantly higher wages for holding their licenses, so they are expected to shoulder the blame when necessary [22]. Such a blame culture does not consider systemic issues or latent failures, discouraging open communication and honest disclosure [5]. Recognizing this reality, numerous entities in commercial aviation, including international organizations like ICAO and regulatory bodies like the FAA, are actively working towards fostering a just culture [21,23,24].

Just culture is "...an atmosphere of trust, where people are encouraged, even rewarded, for providing essential safety-related information—but in which they are also clear about where the line must be drawn between acceptable and unacceptable behavior" [2]. In other terms, just culture means a culture in which front-line operators or others are not punished for actions, omissions, or decisions taken by them that are commensurate with their experience and training but where gross negligence, willful violations, and destructive acts are not tolerated [25]. Another perspective provided by Reason [2] pointed out that just culture refers to "a way of safety thinking that promotes a questioning attitude, is resistant to complacency, is committed to excellence, and fosters both personal accountability and corporate self-regulation in safety matters".

Transitioning from a blame culture to a just culture requires establishing a robust and functional reporting system. Such a system enables individual employees to report their errors or identify systemic hazards without fear of retaliation [5]. The level of trust between employees and management within an organization is the most crucial factor in determining and predicting the success of such a system. A safety culture that is characterized as just encompasses both individual and organizational aspects, extending to attitudes and structures. Personal attitudes and the culture within an organization can either enable or hinder the emergence of trade-offs and operational variabilities [12].

A just culture does not imply unconditional forgiveness for the responsible party in the event of an error [3]. It is founded on the understanding that professionals such as pilots, aircraft mechanics/engineers, and air traffic controllers must adhere to certain fundamental professional and ethical standards. If they operate within these standards and make errors, disciplinary or punitive measures may not be imposed. However, if their performance violates these standards, the error is deemed unacceptable and may result in disciplinary action. The EUROCONTROL, a civil-military organization dedicated to supporting European aviation, has a Just Culture Task Force composed of legal and safety experts of the Member States, the European Commission, air traffic management (ATM), and air transport associations. This group has developed a model for a policy regarding criminal investigation and prosecution of aviation and railway incidents and accidents [26]. This model contributes to achieving a balance between the administration of justice and safety requirements in accident investigations. It shows European aviation's interest in promoting a just culture and maintaining public confidence.

The balance between the administration of justice inside organizations and reporting culture includes a moral issue that Dekker and Breakley [27] addressed by analyzing retributive justice and restorative justice. They argue that accountability considerations must be set alongside deeper concerns with safety and justice. Retributive justice focuses on establishing the appropriate course of action for individuals who have violated specific standards of behavior. Restorative just culture focuses on learning why it made sense to the operator to do what they did, looking ahead at promoting trust to repair the relationship between people whose safety depend on one another. This approach facilitates healing and drives more learning than a retributive approach.

Learning from events is the primary objective of just culture. However, promoting a reporting culture and assessing events effectively is a complex process and brings challenges. Different approaches address the challenges of implementing a just culture [28,29]. In terms of fostering a just culture, it was found that a hierarchical structure that values and promotes a retributive response adds complexity to cultivating a restorative culture. Senior management must effectively address these challenges to foster an environment that prioritizes restorative justice principles [30].

When it comes to modeling safety or just culture, previous findings suggest that several factors, including employees learning from behavioral outcomes, the presence of supportive conditions, and the maintenance of consistency over time, could influence the development of safety culture [18]. Systematic evaluation of safety culture in socio-technical organizations also highlighted the impact of the formal structure of organizations and the values and beliefs that drive individual behavior within organizations [19].

2.3. Modeling: Theoretical Background

Modeling plays a crucial role in the relationship between systems and knowledge. The real world provides actual experiences translated into simple, complicated, or complex phenomena [31]. Models can be used to represent the real world for analysis purposes and support the construction of wholes (synthesis world). The aspect knowledge world is the foundational knowledge that can model phenomena (science) and express abstractions [32]. In the information systems discipline, conceptual modeling is a foundational process for capturing, organizing, and representing the essential elements of a system or domain [33]. At its core, modeling revolves around creating abstract and simplified representations, known as conceptual models, that encapsulate the key concepts, entities, attributes, and relationships relevant to the system under development. Therefore, modeling refers to constructing these conceptual models using various modeling languages, notations, and techniques. This modeling process involves translating real-world complexities into structured and understandable forms, enabling stakeholders to gain insights into the system's structure, behavior, and requirements [33].

Theoretical foundations for conceptual modeling include ontological, epistemological, linguistic, and pragmatic principles [34]. These foundations allow for the creation of conceptual models capable of representing a domain and supporting the solution given a concern [32]. The model's ability to capture knowledge about the relevant domain depends on ontology, semantics, language, and communicative acts [34]. In summary, the conceptual aspects inherent to modeling consist of developing a clear understanding of the domain to be modeled, abstraction, knowledge, and visualization. Visual representations, such as diagrams to depict the conceptual model, ease the communication and understanding of the modeled system. Technical aspects focus on using modeling languages, tools, and techniques to represent, analyze, and simulate the conceptual model [35].

Enterprise architecture modeling employs conceptual modeling techniques to capture and depict the fundamental elements and relationships within an organization's complex system, aiming to provide a holistic view that supports strategic planning, governance, and transformation initiatives. Enterprise architecture modeling is a systematic process of creating abstract representations of an organization's structure, processes, systems, and interactions to facilitate decision-making, alignment, and optimization of its enterprise-wide resources and capabilities [36]. The holism and the synthesis construction of enterprise architectures serve the enterprise (business, systems, and organizations) domain and its inherent concerns. A framework, language, and methodology can define the abstraction level of a conceptual enterprise architecture [37]. More information about EA as a methodology is presented in the next section.

2.4. Safety Management in Aviation

The primary objective of a Safety Management System (SMS) is to establish a structured and comprehensive approach to managing risks and ensuring the effectiveness of safety controls [1]. In a highly regulated industry like aviation, SMS serves as a means for organizations to comply with safety requirements and assess their management capabilities. Besides compliance aspects, SMS enables the integration of safety principles into all aspects of an organization's operations. This integration includes organizational processes, strategic planning, resource allocation, leadership, performance monitoring, and staff recruitment [38].

In order to limit the scope of SMS for this article, we will focus on the components used by the modern aviation industry. In 2006, ICAO proposed a framework for SMS in aviation with four components: safety policy and objectives, safety risk management, safety assurance, and safety promotion [20]. By implementing the principles and requirements outlined in Annex 19, ICAO aims to improve safety performance across the aviation industry. The goal is to enhance safety performance, reduce accidents and incidents, and foster a proactive and systematic approach to managing safety risks. Regulation authorities also rely on the four SMS components. The Federal Aviation Administration (FAA), the European Aviation Safety Agency (EASA), and other regional authorities adopt ICAO standards and develop their regulatory frameworks and guidelines based on Annex 19. They ensure that aviation providers, including airlines, air taxi operators, corporate flight departments, and pilot schools, comply with SMS requirements and integrate safety management into their operations.

Hollnagel [39] explored safety management by applying resilience principles, in which effective safety management requires learning from the past and anticipating the future. Accident investigation (learning from the past) and risk assessment (projecting the future) depend critically on the organizations' models and methods. In his work, the variability of human actions is not a threat but a crucial resource. Therefore, four abilities characterize resilient organizations and are applicable to SMS: (1) the ability to respond to current challenges; (2) the ability to monitor incoming critical situations; (3) the ability to anticipate the occurrence of future events; and (4) the ability to learn from the past [40]. Further, a systematic review evaluated resilience engineering among other theoretical perspectives in the aviation safety context [41]. Although resilience engineering was the dominant approach, the analysis shows that past studies mainly looked at the primary operational aviation subsystems (air traffic control and flight operations). Secondary operational subsystems, such as ground operations and aircraft maintenance, were considerably less used, suggesting a gap in ecosystem perspectives.

Aviation providers have applied and learned from the SMS framework in recent decades. Technological advances in aviation have increased the complexity of operations, and the limitations on SMS have been identified and discussed. A pragmatic approach in [42] analyzed the SMS limitations based on the four components and provided a comprehensive perspective. Another study evaluated the complexities of safety management through a systems thinking lens [20]. It suggests a systemic approach to managing the paradoxical states of the SMS and balancing dynamics within the organization.

2.5. Urban Air Mobility Concept of Operations

Urban Air Mobility (UAM) considers using small aircraft such as drones, air taxis, and other aerial vehicles for transportation in urban and suburban areas. UAM seeks to provide a fast and efficient mode of transport, circumventing ground congestion and, thus, reducing passenger travel times [13]. Stakeholders, including authorities, service providers, communities, and vehicle designers, are actively developing strategies for UAM operations. However, the inherent complexity of multiple entities operating within urban spaces presents significant challenges. These challenges include technical constraints, infrastructure requirements, regulatory considerations, public acceptance issues, and operational hurdles such as safety, security, noise pollution, and environmental impact, all of which pose barriers to successfully implementing UAM operations [43].

The technology nuances and emergent behavior of a complex ecosystem like UAM are an abundant source of analysis, approaches, and exploration toward safe operations. The expansion of UAM into cities introduces a multitude of stakeholders, including multiple operators and service providers, and the establishment of numerous vertiports across the urban landscape. While this distributed operational network brings about various advantages, such as improved accessibility and flexibility, it also presents significant challenges in terms of managing safety [9,15,44]. Ground transportation services are expected to be part of the UAM mobility scenario. Integrating air and ground transportation for passengers adds more providers and organizations to the UAM landscape. Each operator and service provider may have unique operating procedures, aircraft models, and safety protocols, leading to a diverse operating environment that requires careful coordination and harmonization [45].

Macro-level aspects such as urban settings and external factors like regulatory frameworks, public policies, societal attitudes, and economic factors all play crucial roles in shaping the environment in which UAM operates and the culture of safety that prevails. The industry's novelty, urban environment, distributed operational network, and integrated mobility services are macro-level aspects that affect the safety performance and, consequently, the internal aspect and the micro-level behavior of organizations participating in a just culture. The successful integration of UAM into urban environments necessitates a comprehensive understanding of the challenges at the macro- and micro-levels.

By leveraging real data, learning from mistakes, and promoting a culture of safety and accountability, the UAM industry can pave the way for a sustainable operation. Furthermore, the concept of just culture plays a crucial role in generating the necessary information for learning within the UAM industry. It also encourages open communication, transparency, and accountability by creating an environment where individuals can report safety incidents and near-misses without fear of retribution. This enables the collection of valuable data that can be analyzed to identify system weaknesses, develop targeted interventions, and continuously enhance safety practices within the UAM ecosystem.

The UAM Concept of Operations released by the FAA [6,7] delineates the planned operational setting across phases to accommodate the growth of flight operations and airspace traffic. Similarly, the EmbraerX Concept of Operations [8] addresses this issue through a comparable approach, outlining various scenarios involving a growing number of vehicles sharing airspace and an increased utilization of UAM transportation by passengers. UAM phases were then defined according to the horizons projected. Despite the expected evolution towards autonomous flight in UAM operations, this article focuses on the initial stage where pilots maintain control of the vehicle, aiming to provide insights into this critical phase of UAM operations.

While other studies have addressed operational barriers for UAM operations [13] and challenges to UAM safety management [9,15,44], no work has been conducted to address organizational perspectives to foster a just culture in UAM. The results in Section 4 consider the UAM ecosystem perspectives, including operators, service providers, physical infrastructure, authority, and the public. This research elaborates on architectural views addressing the safety management values and perspectives necessary to enable a just

culture. The holistic and strategic proposition using enterprise architecture views represents a unique approach to the questions raised in the introduction.

3. Methods

3.1. Enterprise Architecture Methodology

Modeling methodologies play a significant role in representing a System-of-Systems like the Urban Air Mobility (UAM) ecosystem, providing a comprehensive understanding of its elements and interactions. The challenges of dealing with complex systems can be alleviated by employing modeling techniques, and tools can be utilized to analyze behaviors and relationships. In this context, Enterprise architecture (EA) emerges as a holistic discipline that tackles the intricacies of an enterprise while ensuring its responses align with desired business objectives and outcomes [46]. The objective of adopting enterprise architecture views is to understand its entities and their interactions thoroughly.

An enterprise is a system in which the components are the enterprise's resources. Thus, EA is anchored on system sciences, the discipline that provides the necessary foundations to model and design systems. The contribution of EA regarding the systemic philosophy and paradigm was explored in [47]. Regarding heuristics, enterprise-level models are related to System-of-Systems principles [48]. Furthermore, the core elements of System-of-Systems engineering [49] have been linked to EA processes [50].

EA facilitates enterprise analysis, planning, governance, and evaluation as a welldefined practice. It employs holistic approaches for the successful development and implementation of strategies. Architects can leverage EA to apply architectural principles and practices, guiding organizations through changes in business, information, processes, and technology required to execute their strategies [51]. These practices utilize various aspects of an enterprise to identify, motivate, and implement these changes.

This research employed the enterprise architecture methodology described in the international standard ISO/IEC/IEEE 42020 "Software, systems, and enterprise—Architecture processes" [46]. The standard defines six architecture processes listed below with their intended purpose.

- 1. Architecture Governance: Establish and maintain alignment of architectures (goals, policies, and strategies and with related architectures);
- 2. Architecture Management: Implement architecture governance directives;
- Architecture Conceptualization: Characterize the problem space and determine suitable solutions that address stakeholder concerns, achieve architecture objectives, and meet relevant requirements;
- 4. Architecture Evaluation: Determine the extent to which architectures meet their objectives, address stakeholder concerns, and meet relevant requirements;
- 5. Architecture Elaboration: Describe an architecture in a sufficiently complete and correct manner for the intended uses of the architecture;
- 6. Architecture Enablement: Develop, maintain, and improve the enabling capabilities, services, and resources needed to perform the other processes.

To answer the research questions and meet the intended use of the architecture effort, this article focused on two processes: architecture conceptualization and architecture elaboration. In the first part of the results, we tailored the architecture conceptualization process to characterize the problem space and determine the solution space (workflow described in [46] (pp. 30–34), items 8.4.3 and 8.4.6).

The problem characterization activity was also tailored to address the specific problem of the research. It combines context examination, element harmonization, negative and positive influences and interactions between proposed solutions, and established desired functional and non-functional characteristics that correspond to the stakeholders' concerns. More specifically, the problem space (presented in Section 4.1.1) evaluates and synthesizes information to make reasoned judgments for the potential solution envisioned by the solution space (presented in Section 4.1.2).

The second process applied in this study is the architecture elaboration process. Architectural views and models are captured to address the just culture at UAM and answer the research questions. The level of architectural detail was adjusted to express the architecture's fundamental concepts and properties to the extent necessary for their intended use. In addition, the views are aligned with relevant requirements and design characteristics. The organizational views presented below were chosen to address the critical aspects raised by the problem. The objective is to plan and define the enterprise architecture, ensuring cohesive strategy among organizations and entities of the UAM ecosystem while addressing the goals to overcome the safety culture problem.

3.2. Modeling Framework

The analysis, design, planning, and implementation of EA involve the development of EA models for visualization. The modeling enhances the abstraction degree of information system design, allowing for early-stage verification in system development. EA visualization provides information system designers with a comprehensive view of business and organizational aspects. In large-scale projects, EA visualization facilitates the management of resources, strategies, risks, and business processes [51]. To this end, the EA modeling framework utilized in this article is the Unified Architecture Framework (UAF). Thus, the views generated during the architecture elaboration process were modeled according to the UAF standard. Notably, both enterprise architecture (EA) and the Unified Architecture Framework (UAF) have demonstrated their value in constructing diverse and complementary perspectives of the UAM ecosystem [50,52,53].

The modeling framework incorporates into the EA using the guide [54], the UAF Modeling Language (UAFML) [55], and the UAF Metamodel [56] specifications. UAF adheres to the principles of EA modeling while allowing for flexibility and customization. The viewpoints in the Unified Architecture Framework offer a comprehensive platform for defining various aspects of enterprise operations, encompassing processes, requirements, capabilities, human interactions, personnel organization, and roles and responsibilities. However, it is important to note that this methodology carries a potential drawback as it permits the creation of inconsistent or incoherent architectural elements. To address this concern, validation is strongly advised, especially when collaborating with multidisciplinary teams and stakeholders. Furthermore, tailoring the modeling process is recommended as it can enhance the overall quality of the work.

3.3. Architecture Validation

The validation process was performed internally, using the methodology validation processes and modeling verification, and externally, evaluating the solution proposed by this study in accordance with operational context and safety principles. External validation included professionals in the safety, flight operations, and systems engineering field, as described in Section 1. These two validation efforts (internally and externally) occurred iteratively, recursively, and sometimes concurrently.

The methodology offers guidance for validating architecture as described in [46,57]. The standard ISO/IEC/IEEE 21840:2019(E) [56] defines on page 53 the purpose of the validation process as "to provide objective evidence that the system, when in use, fulfills its business or mission objectives and stakeholder requirements, achieving its intended use in its intended operational environment." Validating the architecture involves modeling verification and validation, as well as the architecture evaluation process defined in [46].

With respect to the modeling effort, the authors performed a consistency check and semantic analysis considering the elaboration process and modeling framework [54]. Although not all views, model aspects, and elements are presented in this work due to space limitations, the enterprise motivation and strategy were defined according to UAM just culture goals. From the defined goals (mission and values), the strategic capabilities are then defined and associated with goals. The strategic capabilities are the base for defining functional architecture, resource architecture, and personnel structure. More explanations

of the elements and the development of the architecture are described in the following section. Thus, verification and validation were conducted through a set of analysis activities including element associations, tracing, and coverage after multiple iterations and corrections. Language verification was also performed to check the semantics.

The stakeholder's validation (external) is essential in evaluating the effectiveness of the proposed architecture. It validates that the architecture aligns with stakeholder requirements and expectations. Validators can verify that the architecture adequately addresses business needs, technical specifications, regulatory compliance, and other key requirements, enhancing overall alignment and stakeholder satisfaction [58]. Experienced practitioners in systems engineering, safety, and pilots were invited to understand the research and the different architectural views. The discussion carried out by the multiple perspective groups has the purpose of challenging the architecture elaboration, the ideas represented by diagrams according to the safety theories, and practicality in the UAM's future operations. Different feedback were received, which included improving the diagram presentation, correcting resource exchange naming and flows, including new elements, and modifying responsibilities. The discussions with multidisciplinary teams help the process of conceptual modeling and, most importantly, the qualitative assessment of the solutions proposed [46].

The diagrams presented in the following sections used UAF version 1.2 [55]. The authors created all the diagrams in this article based on the safety management and just culture theoretical foundations presented in Section 2.2. High-level operational concepts were extracted from existing UAM ConOps [8].

4. Results

4.1. Safety Management Conceptualization

The architecture conceptualization process defined by [46] (p. 27) is "to characterize the problem space and determine suitable solutions that address stakeholder concerns, achieve architecture objectives and meet relevant requirements. (...) Conceptualization is where there is a special focus on identifying solutions, but with also an emphasis on fully understanding the complete problem. This also entails defining and establishing architecture objectives, as well as negotiating with key stakeholders on prioritization of their concerns". The conceptualization process involves many activities and tasks. This section presents the architecture conceptualization process tailored to

- a. understanding the problem from the stakeholders' perspectives and interests;
- b. synthesizing the solution to address just culture in the UAM enterprise.

4.1.1. Problem Space

Defining the problem of safety management enabling just culture in the future UAM operations requires a systems thinking approach. Our first step is looking into different perspectives and understanding the forces that drive the environment. Opposite interests within an organization and with external entities are dynamic and must be understood. The relationship among these entities is part of the problem, and acknowledging the boundaries behind each interest is crucial for achieving a balance. To explore the systems dynamics of safety management in UAM, we organized the analysis into administrative, economic, and operational axes.

The *administrative* axis represents the first aspect (see first column of Figure 2). It concerns the rules and regulations governing licensing, operational procedures, and regulatory compliance. At one end, the regulation authority pushes the organization with rules, inspections, and audits. The objective is to protect the public, ensuring the highest level of safety in aviation operations. It means establishing and enforcing aviation safety regulations, guidelines, and standards. Our study focuses on the utilization phase, where the service will be provided to passengers. The enforcement then relates to air traffic management, maintenance practices, and operational procedures. The undesired state of this gradient is a reckless and negligent operation. In the opposite direction, there is the

perspective of just culture. It is about the movement towards reporting events without fear of blaming. The goal is to learn about events and avoid operational silence, in which the organization is unaware of risks and near misses because operators are not reporting them. The enforcement of the regulation is associated with fear of blaming, which in turn makes operators less willing to report issues [5]. Hence, it is not possible to establish a just culture if the regulatory force is adjusted for punishment purposes only.



Figure 2. Safety management stakeholders and their interests.

In the *economic* axis (second column), a business interest protects the profit and viability of the operation. Most of the time, the top-level management represents this interest, constantly looking for productivity and avoiding a state of low profitability (or nonprofit). Efficiency is pursued, and a strategy to increase profit is usually implemented. In the opposite direction, there is the perspective of a safety campaign that aims to promote safety awareness among workers. In this case, the interest is aligned with the SMS component *safety promotion* and, in practical terms, aims to educate workers to make decisions that would not jeopardize the operation's safety. The slogan "safety first" can directly impact productivity when the operation is delayed, interrupted, or canceled. The balance problem about the operational axis is also called the "dilemma of the two Ps" in which there exist conflicting goals: production (delivery of services) or protection (safety) [21].

The last axis represents the *operational* aspect (third column of Figure 2). The operational processes should be defined to achieve an acceptable workload. If work processes are poorly designed or lack optimization, it can contribute to an excessive workload. Inefficient workflows, unclear task assignments, or redundant steps can all contribute to increased workload demands on workers. Moreover, unrealistic time constraints or tight deadlines can create a high-pressure environment and increase the perceived workload. An unacceptable workload can compromise operational safety and is the undesired state of this force. The other side of this axis is the resource management. Its purpose is to provide sufficient human, system, and training resources. Those gradients can be opposites if a process is defined with unrealistic resources or if the best resource is provided, but the operational process is unacceptable.

The gradients were organized in three axes to ease the identification of dynamic forces that coexist in safety management. However, the gradients are not independent, straightforward, or disconnected. Each one of the forces is influenced by and wields influence over other forces. For instance, safety campaign has a direct relationship with the just culture gradient. The reporting culture results from many other processes and values that the safety campaign promotes. Another example is when the business force manages the budget and controls the expenses, constraining resource availability. Figure 2 introduces opposite forces that compose the problem of managing safety to enable a just culture. It will be used to discuss the perspectives and viewpoints of Section 4.2.

4.1.2. Solution Space

The front-line worker is any operator in direct contact with the operation. The worker can be a maintenance staff member, a traffic controller, a pilot, or any other role performing operational processes. They are also the centerpiece of our proposal. Their presence in daily operations and knowledge about field operations are the primary source for generating valuable information in improving safety. Near misses and even mistakes are symptoms (not causes) that can be investigated and assessed if the front-line worker feels safe to report them [3]. Reporting an event is an act of trust and ownership [5]. There is an individual and collective behavior behind reporting culture. Figure 3 represents the high-level concept of just culture from the front-line worker's perspective.



Figure 3. Front-line worker's perspective in a just culture environment.

The operator must have direct access to a reporting system, and his/her direct manager has no other power influence than to build trust and support the habit of reporting. The fear of being scolded by a superior can discourage reporting. Power relationships must be carefully established to avoid influencing the front-line worker's decision, who is usually vulnerable. High-level management must sponsor safety practices and behavior. Values, especially trust, in the organization depend on the alignment among power, discourse, and actions from top to bottom. Similarly, the team or group that will receive, analyze, and sometimes investigate the event needs direct access to the front-line worker. It is also important that this group is from a separate division and has no relationship with the operator in terms of hierarchy. The engagement of the front-line worker in the assessment of events (and other safety-related processes) is also crucial for building ownership.

Trust is the principal value of this concept. The front-line worker not only trusts that a fair assessment will be based on holistic perspectives rather than blaming individuals, but also trusts that the process of reporting for the purpose of learning is effective. The front-line worker must realize that the information provided in the report is useful and will turn into a learning experience. If the operator is not involved in this process or is not able to see changes and improvements, the reporting becomes only bureaucracy, which is the spending of time without purpose.

The authority is an external entity but has a significant role in the just culture context. Establishing an agreement for disclosure and cooperation between the authority and the safety committee is vital to developing trust as a value among all workers. The goal is to learn, and bringing everyone to the table means broadening perspectives and having the means to perform investigations comprehensively and provide a fair assessment.

4.2. Safety Management Perspectives

The architecture elaboration process defined by [46] (p. 47) aims to "describe an architecture in a sufficiently complete and correct manner for the intended uses of the architecture". The views presented below intend to provide visualizations and support the analysis needed to answer the research questions. In other words, the diagrams presented below include different organizational levels perspectives and provide means to understand the individual and collective contribution to the just culture in the UAM ecosystem.

4.2.1. Drivers

The first step in the model development is to define the drivers. *Drivers* in UAF are factors that have significant impact on the activities and goals of the enterprise. Drivers also relate to the purpose of an enterprise. From the forces (as presented in Figure 2) or stakeholders' interests defined in the problem space, we have identified six drivers:

Regulation Gradient: In the Administrative axis, the regulation must not allow reckless operation;

Just Culture Gradient: In the Administrative axis, the safety culture must avoid reporting trust issues;

Business Gradient: In the Economic axis, the business gradient must protect the viability of the operation;

Safety Campaign Gradient: In the Economic axis, the safety campaign must offer protection from unsafe operation;

Workload Management Gradient: In the Operational axis, the operational processes must keep the workload within acceptable levels;

Resource Management Gradient: In the Operational axis, Resource Management must keep the resources within sufficient levels.

4.2.2. Safety Management Functions and Capabilities

The personnel process view in UAF concerns functions that have to be carried out by organizational resources. The following figures represent how the drivers are implemented in terms of functions, who performs those functions, and how organizational capabilities are achieved.

Figure 4 represents the relationships in the administrative axis. The *Regulation Gradient* (<<Driver>> element on the top left) is realized when the function *Oversee Airworthiness* (<<Function>> on top center) is implemented. A set of subfunctions is defined as composition elements to provide more information about the regulatory act. For instance, *Issue Certification, Provide Policies, Monitor Compliance,* and *Enforce Legislation* (<<Function>> elements at the top) are parts of *Oversee Airworthiness*. Consequently, the *Regulation Authority* (<<Organization>> element below *Oversee Airworthiness*) is the resource capable of performing the functions. Inside the organization level, there are some primary resources represented as parts that are also performers: *Airworthiness Specialist, Accident Investigator,* and *Certification Specialist* (<<Post>> elements linked to Regulation Authority).

The same applies to the opposite force: *Just Culture Gradient* (<<Driver>> in the left bottom of Figure 4). The <<Function>> *Empower Event Reporting* realizes the Just Culture gradient. In this case, although the *UAM Operator* (<<Organization>> element on the right side) is the operational performer of UAM services and the organization behind the function *Empower Event Reporting*, the elements linked with the <<IsCapableToPerform>> dotted line

are its parts. It means that different levels within the *UAM Operator* represented as <<Post>> elements are directly engaged with the reporting function and the subfunctions listed at the bottom of the diagram. The model aims to explore the individual contribution, and every post is capable of performing the reporting function. Safety awareness can be promoted with directive activities for which the safety team is responsible. Yet, promoting trust, accountability, and ownership is also an individual act, and each post should perform it.



Figure 4. Safety management functions from the administrative perspective.

The balance between regulation and just culture forces requires the involvement of the authority and the UAM operator to achieve the capability of *Accountability and Fairness* (<<Capability>> in the middle of the diagram). It means that enabling an environment without fear of reporting is only possible if the regulation enforcement is fair. Moreover, achieving a responsible operation requires accountability and safety awareness. Balancing stakeholders' interests is possible if the enterprise develops capabilities according to the boundaries of each gradient. The <<Capability>> elements on the left side of Figure 5 represent the healthy combination of opposite interests in terms of capabilities. *Accountability and Fairness* capabilities realize opposite drivers: Regulation and Just Culture.

The economic axis balance is represented in Figure 5. In this case, only one organization is responsible for performing functions that realize both drivers: *UAM Operator* (<<Organization>> element in the middle). However, different organizational levels within the *UAM Operator* can perform the functions that will implement *Business Gradient* and *Safety Campaign Gradient* (<<Driver>> elements on the left side). *Top-Level Management* (<<Post>> above the *UAM Operator*) looks after operational viability and is responsible for maintaining profitability. The authority for safety matters, called in this diagram *Chief Safety Officer* (<<Post>> below the *UAM Operator*), is capable of performing the *Promote Safety Awareness* function. Top-level management plans for global strategies, while the

safety team defines safety strategy. A relationship between those organizational levels and functions must be coordinated to achieve *Sustainable Operation* (<<Capability>> in the left middle). The capabilities on the left side of the diagram are also harmonized to balance opposite concerns: *Profitability* realizes the business interest, while *Safety Awareness* realizes the Safety Campaign interest. In the next section, we explore the resource exchange elements between organizational levels that can address complementary functions to achieve a balanced capability.



Figure 5. Safety management functions from the economic perspective.

The last diagram in Figure 6 is for the operational aspect. It represents the balance between defining operational processes considering an acceptable workload and, complementarily, providing enough resources according to the same operational processes. Managing resources and workloads is a highly coupled activity, although different capabilities are being delivered. Defining enough resources depends on process definition. Managing workload also depends on the resource's definition. From the operational perspective, the organization wants to avoid defining processes with a workload that is too high or too low. In both cases, unappropriated workloads can affect the safety performance of a process. At the same time, resources must be provided per the process's workload. Lack of resources (personnel, systems, or training) can increase the workload or even prevent the activity from occurring, compromising its performance. This balance is represented by the capabilities *Acceptable workload* and *Sufficient Resources* (on the left side of Figure 6). The enterprise must realize both capabilities to achieve *Safe Operation* (<<Capability>> in the center).



Figure 6. Safety management functions from the operational perspective.

Achieving complementary capabilities means realizing opposite drivers from the operational axis. The high-level functions that realize those drivers are *Ensure Acceptable Workload When Performing Operational Processes* and *Provide Sufficient Resources* (<<Function>> linked with a dotted arrow to <<Driver>> elements). The list of subfunctions that composes the high-level function provides more details on how the different perspectives can support each other. Matching the functions specified for realizing both drivers is essential and strategic. For instance, functions like *Define Operational Processes According to Resources* (fifth <<Function>> on the right side's list) and *Define Resources According to Operational Processes* (first <<Function>> on the left side's list) can raise the necessary alignment between different interests. The resource capable of performing functions on the operational axis is the *Safety Team* <<Organization>>, which is part of the *UAM Operator* <<Organization>> and has a different hierarchical structure than that of the front-line workers.

The structure within *Safety Team* (<<Post>> elements at its right side) and *UAM Operator* (<<Post>> elements above it) are represented in the diagram as essential parts involved in both functions: *Ensure Acceptable Workload When Performing Operational Process* and *Provide Sufficient Resources* (<<Function>> elements above and below *Safety Team*). The operational axis depends on actual field data to plan, adapt, and improve operations. Moreover, the front-line workforce is a critical resource providing inputs for adjusting workload levels and resource gaps.

To reiterate, the above personnel processes diagrams show the implementation of drivers in terms of functions and capabilities and the individuals or entities responsible for performing those functions. From the operational viewpoint, it is possible to specify which activities will implement each function, including methods, parameters, and measurements. Different posts will perform the same function in different ways. The perspective of each organizational level at contributing with a high-level function and, therefore, a capability is valuable to disseminate ownership and awareness. Knowing how each agent in the ecosystem acts towards a common capability is meaningful for developing a holistic perspective.

Capability is defined in UAF as an enterprise's ability to achieve a desired effect realized through a combination of ways and means (activities and resources). The personnel process views above show how stakeholders with different interests can achieve balanced capabilities that meet the UAM's interest as an enterprise.

4.2.3. Safety Management Structure and Interactions

The personnel structure view concerns the organizational structure used to support capabilities. It shows organizational structures and possible interactions between organizational resources. Once we have defined the capabilities that will balance the drivers and how (functions) will be achieved, we can look into the organization-level perspectives. The following diagrams represent the three main organizational level's responsibilities and resource exchange for exhibiting the capabilities.

The first is the *Top-Level Management* perspective. In our model, the top-level management can be represented as *Executives*, Body of Directors (*BoD*), or *Owner* (all <<Post>> elements in the top left of Figure 7). The diagram below shows a set of responsibilities that the highest level in the organization must have. Ensuring operational viability, allocating funds for safety promotion, developing strategic safety plans, making safety-committed decisions, and facilitating cooperation and collaboration for safety are duties allocated to the highest level (represented as <<Responsibility>> elements). Moreover, workforce perspectives around the organization's governance should not be disregarded. Fostering just culture and being accountable is also a responsibility that the power and discourse of the top-level management must be aligned with.



Figure 7. Top-Level Management Engagement.

The exchanges between the top and other organizational levels are on the diagram's right-top side (represented as dotted lines with <<ResourceExchange>> *RE* notations). *Safety Committee* <<Post>> shall provide safety data and strategy (RE7); *Safety Specialist* <<Post>> feeds technical and specialized safety data (RE19); *Front-Line* level (managers and staff <<Post>> elements) provide their valuable perspective (RE9 and RE10). Without these resource inputs, the top-level management cannot meet its responsibility and con-

tribute the necessary capabilities. The resource exchanges proposed by the model are not desirable interactions but a condition (requirement) to develop the strategic capabilities defined earlier.

The next level explored in Figure 8 is the safety division's perspective. Safety core duties such as defining operational safety guidelines, safety promotion and communication, and resource and workload management are defined according to the capabilities driven by the safety campaign and operational gradients. Regarding the just culture gradient, a focus on managing the reporting system and performing a fair investigation is defined to promote trust and accountability. According to the model language, responsibility *exhibits* capability. Nevertheless, this relationship is not straightforward in real operations. Operational processes can include more than one responsibility; likewise, responsibility can be shared in different processes. Responsibilities like *Ensure Front Staff is Engaged in Safety Decisions* (fourth <<Responsibility>> in the left side of the diagram) can be performed in multiple activities from safety division posts. They will impact all the capabilities, not only *Sustainable Operation*. The representation in the diagrams does not show all <<exhibit="https://www.automation.com">exhibit=



Figure 8. Safety Division Engagement.

There is an external and important connection with the *Regulation Authority* (<<Organization>> in the top left) regarding resource exchange. Without cooperation (RE14) between authority and operator, the consensus of acceptable and unacceptable behavior will hardly exist. Cooperation also includes an agreement for disclosure, discussing new practices, and adapting the existing standards. Top-level management offers extensive support (RE8), and *Front-Line Managers* provide their perspective (RE11). The *Front-Line* *Staff* provides a key resource exchange for the safety division. Feedback, performance data, and work-as-done information (RE12) are essential for all the other activities and responsibilities of the safety division.

The last view represents the front-line perspective in Figure 9. The *Front-Line Staff* (<<Post>> element in the middle) must *Develop Operational Ownership* (third <<Responsibility>> in the left side). *Validate Safety Guidelines* (first <<Responsibility>> on the left side) and *Support Event Assessment* (last <<Responsibility>> in the middle) places the staff inside the safety process and brings them knowledge, voice, and accountability. Reporting inadequacies in workload and resources (<<Responsibility>> elements in the middle) will feed the safety division team with valuable information. The daily operational information from the front-line staff is the safety currency that will drive safety management and provide material for learning. Reporting events and encouraging reporting (<<Responsibility>> elements on the right side) is the responsibility of every person involved in the operation. However, it appears in the front-line perspective because firstly, they are the ones seeing and handling the operation, and secondly, they are also vulnerable in terms of hierarchy and power.



Figure 9. Front-Line Engagement.

The front-line manager has a power over the front-line staff expressed by the relationship <<Command>> (dotted line below <<Post>> *Front Line Managers*). The hierarchy and power can discourage the reporting of events and mistakes. The front-line staff must feel comfortable reporting and discussing events with an external entity like the safety division (as in the *Support Event Assessment* <<Responsibility>>). Moreover, the resource exchange RE17 (at the top right) represents this connection between front-line workers (staff and managers) and the safety team. *Safety Practice Guidance and Partnership* (RE17) is crucial to building ownership, accountability, and trust.

5. Discussion

5.1. Debates and Novelties of UAM Safety Management Perspectives

The architectural views presented in the previous section depicted stakeholders' interest under two schemes: at the enterprise level and the group engagement level. The first type of diagram, personnel processes (Figures 4–6), defines enterprise functions that map to desired enterprise capabilities. Meanwhile, the second type of diagram, personnel structure (Figures 7–9), defines group responsibilities towards all strategic capabilities. The synthesis of operational structure, enterprise drivers and capabilities, and roles and responsibilities provided different perspectives of safety management. Although the enterprise approach of this study is unique, the results and insights can be reflected in other literature. This section will highlight some insights and how they relate to research findings on just culture in aviation.

The first point is the construction of values like accountability and ownership. Accountability is present in all the organization-level perspectives of Section 4. Noticeably, the value of accountability does not have a mistake-owner connotation. Pointing fingers and blaming are barriers to a just culture. Instead, accountability brought by a deep sense of ownership of processes, decisions, and operations is the value a safe operation and just culture seeks [5]. Being accountable in the sense of knowing how safety is a collective commitment and a systemic behavior that depends on each one of the ecosystem pieces. Encouraging and establishing a deep sense of ownership and accountability is crucial for fostering a culture where pilots and front-line staff feel empowered to report safety incidents without fear of blame or reprisal.

The second point concerns trust as a result of establishing accountability and ownership. Having different engagement levels aware of the whole safety effort allows individuals to take their share and behave collectively towards a common goal: safety. The responsibility behind the process for each organizational level was defined, considering drivers and individual perspectives. Front-line staff reporting events depends on values like trust, built from ownership and accountability. This aligns with the survey performed with a group of pilots [58], revealing that fear of reprisal from the employer (airline) emerged as the leading reason for failing to report or under-reporting safety information. The lack of confidence in just culture among airlines highlighted the importance of addressing reprisal concerns and fostering a culture of trust and transparency within aviation organizations to encourage voluntary reporting and enhance safety outcomes. Another piece of literature [59] exposed trust issues among pilots. The perception of reprisals for voluntary reporting and the fear of punitive actions can create barriers to reporting and hinder the development of a just culture.

Another point is the power relationship between front-line workers and their direct management. The reporting system and event investigation and assessments were intentionally proposed in a different hierarchical structure, highlighting the need for direct access to reporting systems without fear of reprisal. The evidence touches upon power dynamics and hierarchies within organizations, which can influence the mentality of reporting and the willingness of front-line workers to report safety events. This resonates with another research finding [60] regarding differences in perceptions of disciplinary measures and accountability among pilots and managers. In a just culture, understanding why procedures were violated should be investigated regardless of the outcome, emphasizing the need for a systemic approach that values accountability and transparency at all levels of the organization. Additionally, the safety division's responsibilities were emphasized in effectively addressing events because they directly impact how front-line workers trust the process. This strategy echoes pilots' concerns about timely feedback and the perception that appropriate actions are not always taken upon reporting safety concerns as revealed in previous surveys [60].

The next consideration is regarding organizational attitude towards fostering a just culture. In the enterprise architecture views presented, engagement groups can be applied to any organization directly interacting with the operation. Front-line workers, top-level management, and safety divisions are common categories of personnel, and their roles and responsibilities are suitable for achieving the enterprise's strategic capabilities. The organization can build those values by planning resource exchange while considering front-line-worker voice, power, discourse alignment, and regulatory cooperation. Similarly, recommendations from previous surveys [60] suggest that airlines should develop clear frameworks for acceptable performance and remedial actions following safety events. When airlines embrace the principles of a just culture and pilots understand the significance of their incident reports, it creates fertile ground for preventing future incidents more effectively [61].

The last point to discuss is adopting enterprise architecture (EA) as a method of elaborating and providing visualizations for the results. Considering that UAM has operational and safety challenges that are being discussed and planned to operate in the future, preparing for safety management is a recommended strategy. Enterprise architecture can facilitate communicating with different stakeholders and converge to shared goals, facilitate architectural analysis and decision-making processes, and support large and complex enterprise transformation changes [62]. The findings of [63], which focused on the strategic value of EA in government agencies, support the EA's role in preparing for safety management in UAM operations. By leveraging EA principles, organizations could effectively communicate with various stakeholders and align efforts toward common goals in different domains. EA is already practiced (and sometimes mandated) among federal organizations and government acquisition processes, providing a communication pattern for better understanding and collaboration [64]. Moreover, EA can facilitate largescale enterprise transformation changes necessary to successfully implement UAM stages aligned with governmental UAM initiatives [7,10]. Finally, in the safety culture concern, the communication factor has been investigated, and evidence shows the positive impact of safety communication on employees' safety performance and safety culture, particularly in high-risk industries [65]. The findings also underscored the need for businesses to focus on safety culture variables, especially in stressful working environments, to improve safety performance. Thus, using EA in safety management strategies can enhance coordination, collaboration, and effectiveness in systematically addressing safety concerns.

5.2. Unified Safety Committee Proposal

Unlike civil aviation, which is usually centralized in big airports, the UAM ground infrastructure will be granulated in various vertiports over urban and suburban areas. The envisioned vertiport network will also have different units in terms of space, topology (on the ground or over buildings), and services (only takeoff and landing pods or with a complete service station for battery charging, maintenance, and parking areas). The distribution will also be significant for providers and operators: different vertiport owners or operators, with diversified staff or third-party workers and different operational processes. This is in addition to multiple UAM operators and different vehicle technologies demanding specific ground services.

Another relevant aspect of the distribution and operational landscape of UAM is the convergence of air and ground transportation sectors. Air travel in dense urban areas would require integrated mobility solutions, including accessibility and connectivity to ground transportation. The air industry is dominant in terms of safety protocols and standards. Many front-line workers, managers, and staff in UAM operations will likely come from aviation backgrounds, where a strong safety culture is typically ingrained. On the other hand, individuals working in ground transportation, such as vertiport operators, ground staff, and service providers in Mobility as a Service (MaaS), may have different levels of recognition or familiarity with aviation safety practices. This difference in safety awareness and training could potentially lead to a mismatch in safety expectations and practices within the UAM ecosystem.

Education and the environment need to be considered to address this issue. First, the UAM industry must establish comprehensive training programs and safety protocols that

bridge the gap between the air and ground transportation sectors. This includes educating ground transportation workers and service providers about aviation safety standards, emergency procedures, and risk mitigation strategies. Collaborative training initiatives and cross-industry knowledge sharing can help align safety practices across all aspects of UAM operations, ensuring a harmonized approach to safety management and enhancing overall safety outcomes in this innovative transportation sector. Secondly, safety management must provide an environment for learning and improving safety performance among all front-line workers.

There is an enormous potential for learning in the new UAM ecosystem. New situations will occur in a systematic and distributed manner. New systems and workers adjusting their performance add another layer of uncertainty to the UAM operations. How can the UAM ecosystem, with all its entities and relationships, learn effectively and efficiently? How can a just culture and all the encouragement for reporting support this learning? The discussion in this section aims to address both questions and propose an environment to enable learning in the UAM ecosystem.

The diagram in Figure 10 shows the main <<Organizations>> involved in the UAM ecosystem: *UAM Operator, Vertiport Operator, UATM Operator, Regulation Authority, Service Providers—Third Party*, and *Professional Association*. Enabling a just culture at the ecosystem level involves multiple organizations. Thus, there is a cultural factor that needs to transcend the boundaries of the organization. The primary base for enabling a just culture is the behavior consensus of what is acceptable and unacceptable. In the model's representation, it means that the *Behavior Consensus* is a competence that operators (organization-level) and the regulation authority must have. The dotted lines with the notation <<RequiresCompetence>> pointing to the <<Competence>> *Behavior Consensus* at the center of Figure 10 illustrate this need. We propose the UAM Safety Committee as a unified organization capable of developing and implementing such competence. The realization arrow pointing to the *UAM Safety Committee* (<<Organization>> at the bottom right) synthesizes the purpose of the strategy discussed in this section.

This committee would have representatives (see <<ResourceRole>> inside the UAM Safety Committee box) from all organizations at different levels, including front-line workers, to ensure their voice and ownership value. The event disclosure (RE56) from the *UAM Safety Committee* to the *Regulation Authority* is part of their cooperation agreement. The internal events that fall under a minor level (upon agreement) should stay in each organization and be handled according to its internal policies. Cases requiring a higher attention level are the ones that are brought to the unified committee. When involved, the *Regulation Authority* should participate in relevant discussions (RE57) towards learning and improving operations.

All regulatory processes like surveillance, inspection, audits, compliance demonstration, and law enforcement (RE36, RE37, RE38, and RE39 in the bottom left) are kept between the *Regulation Authority* and each *Operator*. Boundaries for accountability are essential for the regulation and can coexist with a fair assessment discussed between the authority and safety committee. The operators provide safety events and actively participate in operational discussions (RE58, RE59, and RE60 above the *UAM Safety Committee*). The purpose is to create a systemic view with different perspectives of the problem and contribute to better solutions and improved safety. Participation builds ownership that should occur at every level, from front-line daily discussions to a unified safety committee board.

Third-party workers are usually a challenge for organizations in terms of culture and values. This becomes even more critical with the expected diversity of organizations entering the UAM arena, including small organizations without an aviation background. It is essential to include all providers in safety awareness and accountability (RE44, RE45, and RE47 pointing at the *Service Providers—Third party* box). Likewise, the *Passenger* (<<Person>> in the top) is another part of the operation and should be instructed on safety awareness (RE62 below *Passenger*). Everyone should know they are part of a whole new operation, and its safety is an individual and collective commitment. Lastly, the *Professional*



Figure 10. Unified safety committee proposal.

The ultimate goal of a just culture is learning from events. Information about actual operations, including successful practices, near misses, and hazard events, is *the asset* for UAM safety. Having a shared space for discussing events in such a distributed and diversified operation provides an opportunity for assessing this information. Nonetheless, the proposal is for an ecosystem-level solution, and there are also significant challenges. Achieving a consensus with so many players, including authorities debating how to define safety standards, is arduous. Giving space and a voice to front-line workers is also challenging and requires excellent organization, management, and efficiency. In every environment that expects horizontal and collaborative work, some practices flow smoothly, and others do not fit. The recommendation is to start the consensus discussion early on and evolve with the maturation of the UAM. Avoiding building a just culture could compromise the necessary learning and bring even more challenges for public acceptance in case of accidents.

The unified safety committee proposal was discussed and presented for the UAM ecosystem, which comprehends different organizations and governance systems. The holistic approach is novel and reflects on empirical studies with pilots and managers about the line between acceptable and unacceptable behavior [59]. Differences in perceptions of what is considered acceptable or not were found. Gray areas regarding interpretations, technical and non-technical assessments, and decision-making were highlighted. Finally, insights about the uncertainty among pilots and managers regarding disciplinary measures and accountability underscore the need for a unified approach, as the UAM Safety Committee proposed.

5.3. Limitations

A prominent limitation is the reliance on modeling, which inherently simplifies the complex UAM ecosystem. Models are abstractions that may not fully represent the reality of UAM operations. Therefore, the findings should be interpreted as insights and approximations rather than absolute truths. Like any modeling framework, the results are subject to the assumptions and simplifications inherent in the chosen resource. Moreover, the UAF has a metamodel [56] with a defined ontology and semantics. Although the architect has some freedom to create elements, views, and relationships, most of the modeling effort follows the established rules in the language specification and guides [54,55]. The architect is a limitation as a resource that conceptualizes the architecture. Even skilled architects may lack specific knowledge or present bias that brings limitations to the architecture as a product of human cognitive work.

Moreover, the process of conceptualizing and elaborating architecture is deeply influenced by the architect's subjectivity, a unique and valuable perspective in collaborative efforts. Different individuals or groups of architects can propose different solutions for the same problem, each bringing their own distinct viewpoint. This process involves multidisciplinary teams, discussions, validations, iterations, and recursions, all of which are different parts of the methodology that aim to converge to an elegant solution. Notably, the outcome of the architectural elaboration is typically not rigid and final. Modifications, improvements, and situational tailoring are expected and can occur in a manner that supports the enterprise's goal.

Complexity is another aspect that brings limitations. UAM is a complex and evolving System-of-System. Despite efforts to explore it holistically, the dynamic nature of UAM may result in some aspects being beyond the scope of this article. On top of a complex ecosystem, UAM is a conceptual operation. Data availability relies on conceptual studies and proposals for an imagined future. There are no actual data; only extrapolations and reasonable assumptions. The quality and availability of data related to UAM are also limitations. Lastly, there are interdisciplinary challenges. Integrating multiple disciplines, methods, and subjects, as this work does, is challenging. The interdisciplinary approach introduces complexities that require trade-offs in depth and breadth. Since the ecosystem perspective proposes a whole visualization, there are more perspectives than those developed by this research. Likewise, experts invited to discuss and validate the views do not represent all possible contributors. Consequently, the validation and discussion are limited to the expert's experience and knowledge.

5.4. Next Steps and Recommendations

The following steps in the architecture model include detailing operational processes and performers to address the responsibilities identified in the personnel structure. Resource exchanges can also be detailed and decomposed into processes and data. Future contributions to the research can address the challenges discussed with the unified committee proposal. Moreover, a case study can frame an actual location with known transportation providers, local and regulatory authorities, and known vertiport operators. Exploring the challenges with actual players can mature the discussion and promote benchmark knowledge.

Recommendations for regulators include transparent and standardized reporting and investigation procedures for incidents or near-misses in UAM and advocating for increased collaboration and communication among regulatory bodies, UAM operators, vehicle manufacturers, and Original Equipment Manufacturers (OEM). Furthermore, small organizations, subcontractors, or organizations without aviation backgrounds pose an additional challenge to the ecosystem safety culture. Regulators, authorities, and established organizations in the aviation domain shall collaborate and carefully include these small players in discussing event-reporting standards. Utilizing architectural views, such as the unified safety committee structure, can help communicate the overarching safety culture and operational framework to these smaller entities. While regulators and larger organizations may lead in defining event assessment and investigation protocols, EA can serve as a guide for training, fostering accountability, and promoting engagement across all types of organizations. It provides a holistic view that helps smaller entities understand their role, responsibility, and connections within the broader safety landscape, including

aspects like reporting procedures and engagement with unions/professional associations. Although EA implementation may vary based on organizational size and complexity, its strategic use as a guiding framework can facilitate the integration and alignment of diverse players, including new entrants, in the UAM ecosystem.

Lastly, future implementations of the proposed structure and behavior in the UAM architecture can be performed using the design definition process in ISO/IEC/IEEE 15288 [55]. Additional considerations for the design of System-of-Systems (SoS) processes are defined in ISO/IEC/IEEE 21840 [56]. The organizational structure and processes of the architecture represent requirements for the UAM safety management implementation.

6. Conclusions

In conclusion, this article has examined the foundation for establishing a just culture within Urban Air Mobility (UAM) operations. This study has underscored the importance of a learning-oriented environment where incidents and accidents are analyzed to drive improvements and enhance safety performance. The analysis has delved into the boundaries and challenges of implementing a safety management structure that fosters a just culture throughout the UAM ecosystem. Adopting UAF, organizational viewpoints combining drivers, capabilities, responsibilities, and resource exchange established the approaches and methods for enabling a just culture.

The personnel processes and structure presented in Section 4 addressed the enterprise capabilities and relationships able to manage stakeholders with different interests in UAM safety management. The organizational views also contributed to recognizing the front-line worker's central role in ensuring safe and efficient UAM operations. Building trust with these workers is essential for cultivating a just culture that upholds ownership, accountability, and continuous learning. This study has explored factors influencing trust-building efforts, including active participation in safety discussions, reporting mechanisms and safety event assessments not attached to hierarchical relationships, fair enforcement of safety policies, and a supportive organizational environment. Furthermore, this research has shed light on the individual and collaborative nature of achieving a just culture within the UAM system.

This study highlights the significance of incorporating a just culture mindset and practices in the emerging UAM ecosystem. New technologies and uncertainties bring unintended consequences. Building expertise with transparency can enable learning and collaborative safety management. Section 5 discussed the results and reflections with other literature. The ecosystem-level consensus and cooperation among organizations and regulatory authorities were highlighted. A unified safety committee was proposed as a space for discussion, learning, and improved communication within the UAM community.

In summary, the research contribution includes views combining different levels in the operational hierarchy, functions, responsibilities, and capabilities. Furthermore, we analyzed the challenges of UAM Just Culture and discussed a cooperation proposal to foster learning in the UAM ecosystem. This is a novel contribution to the UAM field, providing a comprehensive visualization balancing different stakeholders' interests and promoting a holistic understanding of the safety culture in UAM operations.

Author Contributions: Conceptualization, R.H.; methodology, R.H. and H.N.; software, R.H.; validation, R.H., P.G. and H.N.; formal analysis, R.H. and H.N.; investigation, R.H. and P.G.; resources, H.N.; data curation, R.H.; writing—original draft preparation, R.H.; writing—review and editing, R.H., P.G. and H.N.; visualization, R.H.; supervision, H.N.; project administration, R.H. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: The data presented in this study are available on request from the corresponding author due to the diagrams presented in this study are a piece of a more complete model.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. ICAO. Annex 19 of the International Civil Aviation Organization (ICAO), 2nd ed.; ICAO: Montreal, QC, Canada, 2016.
- 2. Reason, J. Managing the Risks of Organizational Accidents; Ashgate: Farnham, UK, 1997.
- 3. Dekker, S. The Field Guide to Understanding Human Error; Ashgate Publishing Limited: Farnham, UK, 2006.
- 4. Woods, D.; Dekker, S.; Cook, R.; Johannesen, L.; Sarter, N. Behind Human Error; Ashgate Publishing Limited: Farnham, UK, 2010.
- 5. Dekker, S. Just Culture, Balancing Safety and Accountability; Ashgate Publishing Limited: Farnham, UK, 2007.
- 6. Urban Air Mobility (UAM). Concept of Operations 1.0, Federal Aviation Admin; FAA: Washington, DC, USA, 2020.
- 7. Unmanned Aircraft System (UAS) Urban Traffic Management (UTM). *Concept of Operations 2.0, Federal Aviation Admin;* FAA: Washington, DC, USA, 2022.
- 8. *EmbraerX and Air Service Australia, Urban Air Traffic Management Concept of Operations;* Version 1.0; Air Service Australia and Embraer Business Innovation Center: Canberra, Australia, 2020.
- Bauranov, A.; Rakas, J. Urban air mobility and manned eVTOLs: Safety implications. In Proceedings of the IEEE/AIAA 38th Digital Avionics Systems Conference (DASC), San Diego, CA, USA, 8–12 September 2019; pp. 1–8.
- U-Space ConOps and Architecture, European Union Aviation Safety Agency (EASA), 4th Edition, SESAR-VLD2-03-2020. 20 July 2023. Available online: https://www.sesarju.eu/sites/default/files/documents/reports/U-space%20CONOPS%204th% 20edition.pdf (accessed on 23 October 2023).
- 11. Straubinger, A.; Rothfeld, R.; Shamiyeh, M.; Büchter, K.; Kaiser, J.; Plotner, K. An overview of current research and developments in urban air mobility—Setting the scene for UAM introduction. *J. Air Transp. Manag.* **2020**, *87*, 101852. [CrossRef]
- 12. *A Roadmap to a Just Culture: Enhancing the Safety Environment;* Global Aviation Safety Network Working Group E (GAIN): Alexandria, VA, USA, 2004.
- 13. Rajashekara, K.; Wang, Q.; Matsuse, K. Flying Cars: Challenges and Propulsion Strategies. *IEEE Electrif. Mag.* 2016, 4, 46–57. [CrossRef]
- 14. Jang, S. Incorporating Safety Excellence into Urban Air Mobility (UAM): Insights from Commercial Aviation, Rotorcraft, and Unmanned Aerial Systems (UAS). Senior Honors Theses, Liberty University, Lynchburg, VA, USA, 2024.
- 15. Cokorilo, O. Urban Air Mobility: Safety Challenges. Res. Procedia 2020, 45, 21–29. [CrossRef]
- 16. Scientific Assessment for Urban Air Mobility (UAM). International Forum for Aviation Research. 2023. Available online: https://ifar.aero/attachments/article/57/ifar-scientific-assessment-for-uam.pdf (accessed on 10 May 2023).
- 17. Santos-Reyes, J.; Beard, A.N. A systemic approach to managing safety. J. Loss Prev. Process Ind. 2008, 21, 15–28. [CrossRef]
- Karanikas, N.; Kaspers, S.; Piric, S.; Roelen, A.; Boer, R.; Aalst, R. Measuring Safety in Aviation: Empirical Results about the Relation between Safety Outcomes and Safety Management System Processes, Operational Activities and Demographic Data. In Proceedings of the 7th International Conference on Performance, Safety and Robustness in Complex Systems and Applications, IARIA, Venice, Italy, 23–27 April 2017; pp. 9–16.
- Sharpanskykh, A. A Systemic Approach to Organizational Safety Modeling and Analysis. Int. J. Inf. Syst. Crisis Response Manag. 2012, 4, 42–56. [CrossRef]
- Carter, N.; Mansouri, M. Safety Management Complexity: A Systems Thinking Approach. In Proceedings of the IEEE International Symposium on Systems Engineering (ISSE), Vienna, Austria, 24–26 October 2022; pp. 1–6.
- International Civil Aviation Organization. The Safety Management Manual (SMM)—International Civil Aviation Organisation (ICAO) Doc 9859, 4th ed.; International Civil Aviation Organization: Montréal, QC, Canada, 2018.
- 22. Patankar, M.; Sabin, E. *Human Factors in Aviation*, 2nd ed.; Chapter 4—The Safety Culture, Perspective; Salas, E., Maurino, D., Eds.; Academic Press: Cambridge, MA, USA, 2010; pp. 95–122.
- Key, K.; Hu, P.; Choi, I.; Schroeder, D. Safety Culture Assessment and Continuous Improvement in Aviation: A Literature Review; Report DOT/FAA/AM-23/13; Aviation Safety Office of Aerospace Medicine: Washington, DC, USA, 2023.
- Commission Regulation (EU) No 691/2010. Laying Down a Performance Scheme for Air Navigation Services and Network Functions and Amending. Document 32010R0691 of 29 July 2010. Available online: https://eur-lex.europa.eu/legal-content/ EN/TXT/PDF/?uri=CELEX:32010R0691&qid=1715855142613 (accessed on 3 May 2023).
- 25. Safety Standard of Excellence Safety Management Systems, 3rd ed.; Civil Air Navigation Services Organisation (CANSO): Soesterberg, The Netherlands, 2023.
- Model for a Policy Regarding Criminal Investigation and Prosecution of Aviation and Railway Incidents and Accidents, EURO-CONTROL. Available online: https://www.eurocontrol.int/sites/default/files/2020-11/eurocontrol-aviation-rail-just-culturepolicy.pdf (accessed on 5 April 2023).
- 27. Dekker, S.; Breakey, H. Just culture: Improving safety by achieving substantive, procedural and restorative justice. *Saf. Sci.* **2016**, *85*, 187–193. [CrossRef]
- Kováčová, M.; Licu, A.; Bálint, J. Just Culture—Eleven Steps Implementation Methodology for organisations in civil aviation—JC 11. Transp. Res. Procedia 2019, 43, 104–112. [CrossRef]

- Balk, A.; Stroeve, S.; Bossenbroek, J. Just Culture and Human Factors Training in Ground Service Providers; Report NLR-TR-2010-431; NLR Transport Safety Institute: Amsterdam, The Netherlands, 2011.
- 30. Boskeljon-Horst, L.; Snoek, A.; Van Baarle, E. Learning from the complexities of fostering a restorative just culture in practice within the Royal Netherlands Air Force. *Saf. Sci.* **2023**, *161*, 106074. [CrossRef]
- Grabowski, F.; Strzalka, D. Simple, complicated and complex systems—The brief introduction. In Proceedings of the 2008 Conference on Human System Interactions, Krakow, Poland, 25–27 May 2008; pp. 570–573. [CrossRef]
- Natarajan, S.; Kumar, A.; Chaudhuri, S.R.; Nori, K.; Kasturi, V.; Choppella, V. A Conceptual Model of Systems Engineering. INCOSE Int. Symp. 2018, 28, 1720–1736. [CrossRef]
- 33. Thalheim, B. The Theory of Conceptual Models, the Theory of Conceptual Modelling and Foundations of Conceptual Modelling. In *Handbook of Conceptual Modeling*; Embley, D., Thalheim, B., Eds.; Springer: Berlin/Heidelberg, Germany, 2011. [CrossRef]
- 34. Wand, Y.; Monarchi, D.; Parsons, J.; Woo, C. Theoretical foundations for conceptual modelling in information systems development. *Decis. Support Syst.* **1995**, *15*, 285–304. [CrossRef]
- 35. Olivé, A. Conceptual Modeling of Information Systems; Springer: Berlin/Heidelberg, Germany, 2007.
- Polovina, S.; Von Rosing, M. Using conceptual structures in enterprise architecture to develop a new way of thinking and working for organisations. In *Graph-Based Representation and Reasoning: 23rd International Conference on Conceptual Structures, ICCS* 2018, Edinburgh, UK, 20–22 June 2018; Lecture Notes in Computer Science (10872); Champman, P., Endres, D., Pernelle, N., Eds.; Springer: Berlin/Heidelberg, Germany, 2018; pp. 176–190.
- Alexa, S.; Repa, V. Holistic Layer of the Enterprise Architecture on the Basis of Process-Driven Organization. Complex Syst. Inform. Model. Q. 2017, 11, 69–84. [CrossRef]
- 38. Easter, K.; Hegney, R.; Taylor, G. Enhancing Occupational Safety & Health; Butterworth-Heinemann: Oxford, UK, 2004.
- 39. Hollnagel, E. From protection to resilience: Changing views on how to achieve safety. In Proceedings of the 8th International Aviation Psychology Symposium, Sydney, Australia, 8–11 April 2008.
- 40. Hollnagel, E. The four cornerstones of resilience engineering. In *Resilience Engineering Perspectives, Volume 2: Preparation and Restoration;* Nemeth, C.P., Hollnagel, E., Dekker, S., Eds.; Ashgate: Aldershot, UK, 2009; pp. 117–134.
- 41. Muecklich, N.; Sikora, I.; Paraskevas, A.; Padhra, A. Safety and reliability in aviation—A systematic scoping review of normal accident theory, high-reliability theory, and resilience engineering in aviation. *Saf. Sci.* **2003**, *162*, 106097. [CrossRef]
- 42. Malakis, S.; Kontogiannis, T.; Smoker, A. A pragmatic approach to the limitations of safety management systems in aviation. *Saf. Sci.* **2023**, *166*, 106215. [CrossRef]
- 43. Cohen, A.; Shaheen, S.; Farrar, E. Urban Air Mobility: History, Ecosystem, Market Potential, and Challenges. *IEEE Trans. Intell. Transp. Syst.* **2021**, 22, 6074–6087. [CrossRef]
- 44. Wang, L.; Deng, X.; Gui, J.; Jiang, P.; Zeng, F.; Wan, S. A review of Urban Air Mobility-enabled Intelligent Transportation Systems: Mechanisms, applications and challenges. *J. Syst. Archit.* **2023**, *141*, 102902. [CrossRef]
- Ferrão, I.; Espes, D.; Dezan, C.; Branco, K. Security and Safety Concerns in Air Taxis: A Systematic Literature Review. Sensors 2022, 22, 6875. [CrossRef] [PubMed]
- 46. ISO/IEC/IEEE 42020; Software, Systems and Enterprise—Architecture Processes. ISO: Geneva, Switzerland, 2019.
- Wegmann, A. The Systemic Enterprise Architecture Methodology (SEAM) Business and IT Alignment for Competitiveness. In Proceedings of the 5th International Conference on Enterprise Information Systems, Angers, France, 22–26 April 2003; pp. 483–490.
- 48. Brook, P. The Role of Architecting in Systems of Systems. NATO–STO 2015, STO-EN-SCI-276, pp. 1–18. Available online: https://www.sto.nato.int/publications/_layouts/mobile/view.aspx?List=44a8f49d-e481-458a-91b4-212a9605bd9e& View=b927897e-9dc2-4392-aa25-598b0c04b48e&RootFolder=/publications/STO%20Educational%20Notes/STO-EN-SCI-276& ViewMode=Detail (accessed on 28 April 2024).
- Department of Defense (DoD)—ODUSD (A&T), Systems and Software Engineering (SSE). Systems Engineering Guide for Systems of Systems; Version 1.0; Office of the Under Secretary of Defense: Washington, DC, USA, 2008.
- 50. Rubio, C.G.; Rigaut, T. An Urban Air Mobility System of Systems UAF & MDAO Application Case. In Proceedings of the 18th Annual System of Systems Engineering Conference (SoSe), Lille, France, 14–16 June 2023; pp. 1–8.
- Zhou, Z.; Zhi, Q.; Morisaki, S.; Yamamoto, S. A Systematic Literature Review on Enterprise Architecture Visualization Methodologies. *IEEE Access* 2020, *8*, 96404–96427. [CrossRef]
- 52. Hoffmann, R.; Nishimura, H.; Latini, R. Urban Air Mobility Situation Awareness from Enterprise Architecture Perspectives. *IEEE Open J. Syst. Eng.* **2023**, *1*, 12–25. [CrossRef]
- 53. Hoffmann, R.; Pereira, D.; Nishimura, H. Security Viewpoint and Resilient Performance in the Urban Air Mobility Operation. *IEEE Open J. Syst. Eng.* **2023**, *1*, 123–138. [CrossRef]
- 54. Enterprise Architecture Guide for UAF—Object Management Group Unified Architecture Framework (OMG UAF)—Appendix C; Version 1.2; The Object Management Group: Milford, MA, USA, 2022.
- 55. Unified Architecture Framework Modeling Language (UAFML) Version 1.2; Object Management Group Unified Architecture Framework (OMG UAF): Milford, MA, USA, 2022.
- 56. *ISO/IEC 19540-1;* Information Technology—Object Management Group Unified Architecture Framework (OMG UAF)—Part 1: Domain Metamodel (DMM). ISO: Geneva, Switzerland, 2022.

- 57. McMurtrie, K.; Molesworth, B. The Impact of a Legally Defined Just Culture on Voluntary Reporting of Safety Information. *Aviat. Psychol. Appl. Hum. Factors* **2021**, *11*, 88–97. [CrossRef]
- ISO/IEC/IEEE 21840:2019(E); Guidelines for the Utilization of ISO/IEC/IEEE 15288 in the Context of Systems (SoS). ISO: Geneva, Switzerland, 2019; pp. 1–68.
- 59. McMurtrie, K.; Molesworth, B. Confidence and Trust in the 'Just Culture' Construct. *Transp. Res. Procedia* 2022, *66*, 214–225. [CrossRef]
- 60. Schuit, I.; Scott, S. Perceptions of Just Culture Between Pilots and Managers: Evaluation of Airlines in the EU, Middle East, and Asia/Pacific Regions. *Aviat. Psychol. Appl. Hum. Factors* **2021**, *11*, 65–78. [CrossRef]
- 61. Sieberichs, S.; Kluge, A. How Just Culture and Personal Goals Moderate the Positive Relation between Commercial Pilots' Safety Citizenship Behavior and Voluntary Incident Reporting. *Safety* **2021**, *7*, 59. [CrossRef]
- 62. Guo, H.; Gao, S. Enterprise Architectures in E-Governments Studies: Why, What and How? In *Responsible Design, Implementation and Use of Information and Communication Technology (I3E 2020)*; Lecture Notes in Computer Science; Hattingh, M., Matthee, M., Smuts, H., Pappas, I., Dwivedi, Y.K., Mäntymäki, M., Eds.; Springer: Cham, Switzerland, 2020; Volume 12067. [CrossRef]
- Sundberg, L.; Floren, H.; Sundberg, H. Enterprise architecture adoption in government: A public value perspective. In Proceedings of the 16th International Conference on Theory and Practice of Electronic Governance (ICEGOV '23), Belo Horizonte, Brazil, 26–29 September 2023; Association for Computing Machinery: New York, NY, USA, 2023; pp. 254–262. [CrossRef]
- 64. Heikkilä, J.; Penttinen, K. Overview of Enterprise Architecture Work in 15 Countries—Finnish Enterprise Architecture Research Project; Ministry of Finance: Helsinki, Finland, 2007; ISBN 978-951-804-776-9.
- 65. Zara, J.; Nordin, S.M.; Isha, A.S.N. Influence of communication determinants on safety commitment in a high-risk workplace: A systematic literature review of four communication dimensions. *Front Public Health* **2023**, *11*, 1225995. [CrossRef] [PubMed] [PubMed Central]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.